Author's response to reviews

Title: Self-management for osteoarthritis of the knee: Does mode of delivery influence outcome?

Authors:

Sophie Coleman (sophiec@iinet.net.au)
Jean McQuade (jeanm@arthritiswa.org.au)
Jessica Rose (jessrose@iinet.net.au)
Charles Inderjeeth (Charles.Inderjeeth@health.wa.gov.au)
Graeme Carroll (md@arthrocare.com.au)
N.Kathryn Briffa (K.Briffa@curtin.edu.au)

Version: 5 Date: 22 March 2010

Author's response to reviews:

I have made minor typo changes to the abstract and have changed the tables and figure titles in the manuscript because they were incorrectly formatted with both titles appearing on the study design figure.